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Data Taking
Day Delivered Recorded Efficiency Comment

(%)

4 Dec 4.76 3.87 81

5 Dec 6.19 4.94 80

6 Dec 3.53 2.34 66

7 Dec 3.37 2.95 88

8 Dec 4.00 3.39 85

9 Dec 0.48 0.41 86

10 Dec 2.36 2.19 93

* Luminosity data from store 5115 missing

3-9 Dec 24.7 20.1 81

Lum (pb-1) Lum (pb-1)
One hour down for L1muon.  MDT LVPS dies 

at start of store due to high losses. 
Special runs and some problems with L2.  Access 

for MDT and L1muon. Silicon chiller trip.
75 min down at start of store for L1muon.  

Access to replace L1muon card and fix PDT.
Silicon LVPS failure. Special runs taken. Access to 

replace Silicon PS and swap L1muon card.
15 min down for L2 crate.  20 min down for 

STT crate.
Access to fix L1muon card.  Luminosity scaler 

crate hung first 20 min. of store 5115.
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Accesses and Other Events
● D0 requested three accesses this week.

– Muon MDT (2 + 2 hours) – We needed to replace a LVPS.

● This turned off 5% of our forward muon channels.
● Following the access, the silicon chiller and power tripped off.  A pump used in 

measuring the dew point seized and was replaced.  Thanks to AD for additional 
time to diagnose this problem.

– L1Muon card (1 hour) – the L1Muon card that was put in during the previous access 
failed at the start of the next store.

● This failure effected 20% of our forward muon triggers.

– Silicon (2 + 5 hours) – We needed to replace a failed LVPS.

● This power supply powered 1/6th of the silicon.
● After replacing the LVPS, we still saw low voltages, so the supply was replaced 

again.  At this point we also had monitoring problems that kept us from diagnosing 
the correct issue.  A bad rack monitor was replaced along with the LVPS.

– Once again, thanks to AD for providing the additional access time and my 
apologies to the Tev group for taking up all of their study time.
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